Author's response to reviews

Title: Delayed Diagnosis of Intermittent Mesenteroaxial Volvulus of the Stomach by Computed Tomography: A Case Report

Authors:

Colin YL Woon (wolv23@gmail.com)
Alexander YF Chung (alexander.chung.y.f@sgh.com.sg)
Albert SC Low (gdrsc@sgh.com.sg)
Wai-Keong Wong (gsuwwk@sgh.com.sg)

Version: 3 Date: 3 February 2008

Author's response to reviews: see over
Dear Sir / Mdm,

We have much pleasure in sending you the attached revised manuscript for your kind consideration for publication in the JOURNAL OF MEDICAL CASE REPORTS.

<table>
<thead>
<tr>
<th>SN</th>
<th>Item</th>
<th>Change / Remarks</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>Keywords and running head</td>
<td>These have been removed.</td>
</tr>
<tr>
<td>2</td>
<td>Restructure the abstract</td>
<td>The abstract has been structured to include Background, Case Presentation and Conclusions.</td>
</tr>
<tr>
<td>3</td>
<td>Inclusion of an Introduction</td>
<td>An Introduction section has been included in the main body of text.</td>
</tr>
<tr>
<td>4</td>
<td>Three figure maximum</td>
<td>Four figures are included in the original submission. We feel that all 4 figures contribute to the discussion. However, we will be happy to remove figures / merge them into a panel if the editorial board / reviewers feel it is appropriate.</td>
</tr>
</tbody>
</table>
Yours truly,

Drs Colin Woon, Alexander Chung, Albert Low and Wai-Keong Wong

Department of Surgery

Singapore General Hospital

Singapore 169608